Author’s response to reviews

Title: Nothing about me without me: Why an EU health literacy strategy embracing the role of citizens and patients is needed

Authors:

Alexander Roediger (alexander_roediger@merck.com)
Kaisa Immonen-Charalambous (kaisa.immonen@eu-patient.eu)
Markus Kujawa (markus.kujawa@cpme.eu)
Kristine Sørensen (contact@globalhealthliteracyacademy.org)

Version: 2 Date: 28 Feb 2019

Author’s response to reviews:

Answer to comment 1: We have changed the structure as the reviewer has suggested (can be seen in the track changes).

Answer to comment 2: We have changed the sentence in the abstract and the main text accordingly.

Answer to comment 3: We have changed the sentence accordingly and focused on the underestimation.

Answer to comment 4: Done

Answer to comment 5: Done

Answer to comment 6: Done
Answer to comment 7: We have added wording around the concept of "citoyen". We fully agree that this is important and thank you for your support.